Author’s response to reviews

Title: Immunomodulatory factors gene polymorphisms in chronic periodontitis: An Overview

Authors:
Zahra Heidari (histology_iri@yahoo.com)
Bita Moudi (bita.moudi@yahoo.com)
Hamidreza Mahmoudzadeh-Sagheb (histology@ymail.com)

Version: 2 Date: 25 Sep 2018

Author’s response to reviews:

Dear Editor in-chief

Thanks for the kind comments regarding our manuscript. requested modifications were done according to Editor’ points of view. Some sentences were revised in the main-text. We hope that this revision of manuscript could qualify it to be published in the” BMC Oral Healthe”.

Yours’ sincerely

Bita Moudi